Author's response to reviews

Title: Modifiable risk factors for falling and their associations with visiting the Accident & Emergency Department after a fall

Authors:

Alice C. Scheffer (a.c.scheffer@amc.nl)
Pieter Boele van Hensbroek (p.boelevanhensbroek@amc.nl)
Nynke van Dijk (n.vandijk@amc.nl)
Jan S.K. Luitse (j.s.luitse@amc.nl)
Johannes C. Goslings (j.c.goslings@amc.nl)
René H. Luigies (r.luigies@eusamed.eu)
Sophia E. de Rooij (s.e.derooij@amc.nl)

Version: 6 Date: 28 May 2013

Author's response to reviews: see over
Dear Editor,

Thank you for reviewing our manuscript entitled ‘Modifiable risk factors for falling and their associations with visiting the Accident & Emergency Department after a fall’. We are thankful for the opportunity to resubmit this article for BMC Health Services Research. We appreciate the positive approach of the editor reviewers and the useful and valuable suggestions for improvement. We have carefully studied the comments, suggestions and requests.

On behalf of the other authors, yours sincerely,

Alice Scheffer

**Corresponding author**
Alice Scheffer, RN, PhD
Department of Internal Medicine
Section of Geriatrics, Room F4-108
Academic Medical Center,
University of Amsterdam,
Meibergdreef 9
1105 AZ
Amsterdam, the Netherlands
0031 20 5665991 (Phone)
0031 20 5669325 (Fax)
Reviewer report (Eva Ekall Hansson):
We thank the reviewer for her remarks about the revised manuscript
Q1: You need to add a reference in the introduction, second paragraph after the sentence "In this population, secondary fall prevention and primary fracture prevention should be conducted". You need to verify this sentence with one or two references.
A: We have added a reference to the mentioned sentence in the second paragraph of the introduction.

Reviewer report (Irene Vikman):
We thank the reviewer for her positive remarks and her advice to accept the manuscript.

Reviewer report (Andrew D Blann):
We are grateful for the positive remarks made by this reviewer.